Author’s response to reviews

Title: Ethnic disparity in 21-hydroxylase gene mutations identified in Pakistani congenital adrenal hyperplasia patients

Authors:

Aysha Habib (aysha.habib@aku.edu)
Muniba Aban (muniba.aban@aku.edu)
Jamal Raza (jamalraza@yahoo.com)
Naeem ul Haq (rnhaque@gmail.com)
Abdul Jabbar (abdul.jabbar@aku.edu)
Tariq Moatter (tariq.moatter@aku.edu)

Version: 7 Date: 11 February 2011

Author’s response to reviews:

Dear Sir,
Following changes are made in the manuscript as per your recommendations;

1. Authors list is corrected.
2. Precis is incorporated in the main body of the manuscript.
3. Table 2 is cited in the text.
4. Reference # 20 is cited in the text.
5. Spell checked twice to clear typographical errors.

I am sorry for the omissions and hope that the changes made address your concerns. Thank you again

Best Regards,

Dr. Tariq Moatter